Author’s response to reviews

Title: Evaluation of eye-related parameters and adverse events of rigid gas permeable contact lens and spectacles correction in infants with monocular aphakia after congenital cataract surgery: a retrospective clinical study

Authors:

Junjue Chen (787391961@qq.com)
Ping Sun (15216632550@139.com)
Yan Wei (15980532363@139.com)
Xiaoli Kang (Kangxiaoli@xinhuamed.com.cn;zpcjj@163.com)

Version: 2 Date: 30 Jan 2019

Author’s response to reviews:

We have revised the conclusion about spectacles according to your suggestion (Conclusion section in Abstract and in the paper); again, we have discussed the limitations of our article based on your suggestions (the last paragraph of the Discussion section). Thank you very much for taking the time to review on our article and your advice.